## University of Montana

# ScholarWorks at University of Montana

Montana Climate Office Publications

Montana Climate Office

12-2019

# The Topographic Signature of Ecosystem Climate Sensitivity in the Western United States

Zachary Hoylman

Kelsey Jencso

Jia Hu

Zachary A. Holden

Brady W. Allred

See next page for additional authors

Follow this and additional works at: https://scholarworks.umt.edu/climate\_pubs Let us know how access to this document benefits you.

## Authors

Zachary Hoylman, Kelsey Jencso, Jia Hu, Zachary A. Holden, Brady W. Allred, Solomon Z. Dobrowski, Nathaniel Robinson, Justin T. Martin, David L.R. Affleck, and Carl Seielstad

P



# **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1029/2019GL085546

#### Key Points:

- Convergence of hillslope topography contributes to higher ecosystem net primary productivity across the majority of the western United States
- Hillslope convergence reduces ecosystem sensitivity to climate, especially within semiarid regions
- Ecosystems in flat, arid environments are the most vulnerable to drought-induced reductions in productivity

Supporting Information:

Supporting Information S1

#### Correspondence to:

Z. H. Hoylman, zhoylman@gmail.com

#### Citation:

Hoylman, Z. H., Jencso, K. G., Hu, J., Holden, Z. A., Allred, B., Dobrowski, S., et al. (2019). The topographic signature of ecosystem climate sensitivity in the western United States. *Geophysical Research Letters*, *46*, 14,508–14,520. https://doi.org/10.1029/2019GL085546

Received 8 OCT 2019 Accepted 7 DEC 2019 Accepted article online 10 DEC 2019 Published online 23 DEC 2019

© 2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# The Topographic Signature of Ecosystem Climate Sensitivity in the Western United States

Zachary H. Hoylman<sup>1,2</sup>, Kelsey G. Jencso<sup>1,2</sup>, Jia Hu<sup>3</sup>, Zachary A. Holden<sup>4</sup>, Brady Allred<sup>2,5</sup>, Solomon Dobrowski<sup>2</sup>, Nathaniel Robinson<sup>5</sup>, Justin T. Martin<sup>6</sup>, David Affleck<sup>2</sup>, and Carl Seielstad<sup>2,7</sup>

<sup>1</sup>Montana Climate Office, W.A. Franke College of Forestry and Conservation, University of Montana, Missoula, MT, USA, <sup>2</sup>W.A. Franke College of Forestry and Conservation, University of Montana, Missoula, MT, USA, <sup>3</sup>School of Natural Resources and the Environment, University of Arizona, Tucson, AZ, USA, <sup>4</sup>USDA Forest Service, Northern Region, Missoula, MT, USA, <sup>5</sup>Numerical Terradynamic Simulation Group, W.A. Franke College of Forestry and Conservation, University of Montana, Missoula, MT, USA, <sup>6</sup>Department of Ecology, Montana State University, Bozeman, MT, USA, <sup>7</sup>National Center for Landscape Fire Analysis, W.A. Franke College of Forestry and Conservation, University of Montana, Missoula, MT, USA

**Abstract** It has been suggested that hillslope topography can produce hydrologic refugia, sites where ecosystem productivity is relatively insensitive to climate variation. However, the ecological impacts and spatial distribution of these sites are poorly resolved across gradients in climate. We quantified the response of ecosystem net primary productivity to changes in the annual climatic water balance for 30 years using pixel-specific linear regression (30-m resolution) across the western United States. The standardized slopes of these models represent ecosystem climate sensitivity and provide a means to identify drought-resistant ecosystems. Productive and resistant ecosystems were most frequent in convergent hillslope positions, especially in semiarid climates. Ecosystems in divergent positions. This topographic effect was significantly dampened in hygric and xeric climates. In aggregate, spatial patterns of ecosystem sensitivity can be implemented for regional planning to maximize conservation in landscapes more resistant to perturbations.

**Plain Language Summary:** It is well known that gradients in elevation and aspect can have a significant influence on the degree of water and energy available for plant growth and the sensitivity of ecosystems to wet or dry time periods. Little work has examined how hillslope topography and downslope movement of water to zones of convergent terrain can impact plant available water and vegetation growth. We quantified ecosystem response to the climatic water balance (ecosystem sensitivity) across a 30-year record and at a 30-m resolution across the western United States. Our results show that vegetation in zones of hillslope convergence, where moisture from upslope tends to accumulate, is less sensitive to droughts, especially in semiarid settings. Divergent hillslope positions were moderately sensitive to climate and less productive relative to convergent positions. Ecosystem response to topography was dampened in especially wet or dry climates due to significant moisture surplus or moisture deficit, respectively. These distributed measurements of ecosystem sensitivity are important considerations when describing local ecosystem-climate relationships and for identifying management priorities across landscapes. Zones of resistant vegetation are more likely to persist through future droughts, influencing the greater ecosystem's response to climate change.

### 1. Introduction

Identifying the sensitivity of ecosystems to climate is an important consideration in conservation planning (Keppel et al., 2015) and may provide a means to promote climate adaptation (Morelli et al., 2016). Areas of reduced ecosystem sensitivity to climate are likely to support the persistence of climate refugia (Ashcroft, 2010; Dobrowski, 2011; Rull, 2009) and influence climatic connectivity as species move in response to increasing dryness (Carroll et al., 2018; Dobrowski et al., 2013; Dobrowski & Parks, 2016). Therefore, a suggested strategy for conservation and management efforts is to identify factors that contribute to the sensitivity of vegetation to drought and longer term climatic change in order to leverage natural ecosystem resistance (Keppel et al., 2015; Morelli et al., 2016).

Research across the globe has documented productivity-climate relationships associated with regional differences in variables such as temperature, precipitation, radiation, and cloud cover (Churkina & Running, 1998; Schuur, 2003; Seddon et al., 2016; Vicente-Serrano et al., 2013; Vicente-Serrano et al., 2018). However, the sensitivity of ecosystem net primary productivity (NPP) to climate within a given region may also be explained by complex topographic gradients that contribute to lateral flows of energy and water, but at hillslope scales. In many semiarid ecosystems, productivity is constrained by subsurface moisture availability to plants, which is determined by the partitioning of regional climate into runoff, evapotranspiration (actual evapotranspiration [AET]), and subsurface storage (Eagleson, 1978; Rodriguez-Iturbe, 2000). Topographically driven hydrologic processes (e.g., lateral flows of soil water and groundwater; Dingman, 2015) and atmospheric processes (e.g., cold air drainage, vapor pressure deficit-soil moisture feedbacks, and radiation shading; Entekhabi et al., 1996; Frey et al., 2016) modify climatic partitioning of moisture across complex landscapes, producing heterogeneous patterns of plant available water.

Striking differences in subsurface moisture content and atmospheric vapor pressure can emerge across distances of meters due to topographic convergence and divergence in the landscape (e.g., topoclimates/microclimates; Hoylman et al., 2019). Additionally, the hydraulic properties of soil resulting from physical and geochemical soil processes (Brubaker et al., 1993; Seibert et al., 2007), weathering of parent material (Clair et al., 2015; Gabet et al., 2006), and indirect biophysical feedbacks (Dietrich & Perron, 2006; Yetemen et al., 2010) can also influence the storage and release of water to vegetation between upslope and downslope positions. Therefore, at the scale of hillslopes, vegetation can experience a large range of moisture availability and microclimates, which deviate significantly from regional climate conditions due to topographic position.

Vegetation responds to hillslope scale variations in moisture (Hoylman et al., 2018; Istanbulluoglu et al., 2008; Ivanov et al., 2008; Swetnam et al., 2017), creating zones where ecosystems may be buffered from climatic moisture limitations. The degree of buffering is determined collectively by the topographic properties of the landscape (flat vs. convergent, single drainage vs. highly dissected), seasonal delivery of water and energy (climatic water and energy balance), and the physiological responses by vegetation within the ecosystem itself. Areas of reduced ecosystem-climate sensitivity may only exist where regional climatic conditions promote the occurrence of requisite hydrological and meteorological processes (e.g., microclimates, enhanced soil moisture, and near surface groundwater tables) that facilitate establishment, growth, and productivity (McLaughlin et al., 2017).

Here we examine the spatial expression of ecosystem climate sensitivity across the western United States. To characterize ecosystem sensitivity, we computed pixel-specific linear regressions (30-m pixel resolution) between the climatic water balance (deficit = potential evapotranspiration [PET] - AET; Stephenson, 1998) and Landsat-derived ecosystem NPP (Robinson et al., 2018) using 30 years of record (1986–2015). The standardized slope of this relationship represents ecosystem sensitivity and is a simple metric that considers both annual climatic water availability and the resulting plant growth. To isolate the effect of topography, we quantified distributions of ecosystem sensitivity within intervals of the local climatic water balance and hillslope topography (from divergent to flat to convergent). Our results highlight the importance of topography for hillslope scale variations in moisture, creating zones where ecosystems are buffered from climate. Together, these analyses provide important information for the conservation and management of ecosystems from hillslope to regional scales.

#### 2. Materials and Methods

#### 2.1. Ecosystem Productivity Data

We utilized gridded datasets of annual NPP that were produced for the continental United States at a 30-m grid resolution from 1986 to 2015 (Robinson et al., 2018; Figures 1a and 2a and 2b). NPP was calculated using the Moderate Resolution Imaging Spectroradiometer (MODIS) MOD17 algorithm (Running et al., 2004; Sims et al., 2008; Smith et al., 2016). Gross primary productivity was calculated using the daily fraction of photosynthetically active radiation estimates from Landsat surface reflectance products (Feng et al., 2012; Masek et al., 2006; Vermote et al., 2016), meteorological data from gridMET (Abatzoglou, 2013), land cover classifications from the National Land Cover Database (NLCD; Homer et al., 2007; Homer et al., 2015; Fry et al., 2011), and optimized biome-specific parameters (Robinson et al., 2018). NPP was calculated

# **Geophysical Research Letters**



**Figure 1.** (a) The average annual net primary productivity (NPP) and (b) climatic water deficit (deficit = potential evapotranspiration – actual evapotranspiration) for the western United States calculated using annual data from 1986 to 2015. An example region from Montana shows (c) the topographic position index (TPI) of a mountainous region. The negative TPI values represent convergent topographic positions and positive values represent divergent positions. The white regions represent agricultural, urban, and nonvegetated lands that were removed from analysis. (d) The relationship between the climatic water deficit and ecosystem NPP for the western United States. We discretized the entire western United States into 100 quantiles of the deficit. For each deficit quantile the median (black line) and 5th and 95th quantiles (grey ribbon) of NPP are shown. The effect of topography was quantified by repeating this analysis on discrete deficit regions defined by the TPI (lines color coded by red-blue color gradient). The lines transitioning from red to yellow to blue indicate a gradient from divergent to flat to convergent landscape positions. The 5th and 95th deficit quantiles associated with each plant functional type are represented by the horizontal bar and the median value is represented by the scatter point on each horizontal bar.

as the sum of the daily differences between GPP and maintenance respiration, minus annual growth respiration.

#### 2.2. Climatic Water Balance Data

The climatic water balance imparts a strong control over the spatial distribution of plant functional types and is an important driver of ecosystem productivity. We used daily estimates of PET and AET output from a 8 arc-second (~250 m) gridded soil water balance model, evaluated from 1986 to 2015 (Holden et al., 2019), to calculate the climatic water deficit (deficit = PET – AET). We then calculated the total annual deficit for each year, which represents the unmet atmospheric demand for moisture. Using the nearest neighbor method, we resampled the 250-m deficit dataset to match the NPP grid resolution (30-m) in Google Earth Engine (GEE; Gorelick et al., 2017). This climatic water balance is topographically resolved with respect



Figure 2. (a) Diagram showing how pixel-specific regressions are computed across the spatial grid. (b) Scatter plots showing example regressions between the climatic water deficit (deficit = potential evapotranspiration - actual evapotranspiration) and annual net primary productivity (NPP) for three 30-m pixels across 30 years. (c) The spatial distribution of regression slopes for each pixel across the western United States. (d) The spatial distribution of standardized regression slopes (β\*, ecosystem sensitivity) computed using the pixel-specific deficit and NPP distributions. Ecosystem sensitivity accounts for differences in NPP and deficit variance to compare the response of NPP to annual climate dynamics across ecosystems. Negative ecosystem sensitivity values represent ecosystems where NPP declines during times of water limitation, while values close to zero represent insensitive ecosystems. Positive sensitivity values are representative of ecosystems that show increases in NPP during years with increased climatic water deficits and higher PET.

to elevation effects on precipitation, aspect effects on incident radiation, and shading effects from adjacent terrain but does not incorporate lateral flow of water in the subsurface.

#### 2.3. Landscape Topography Characterization

Surface topography has been identified as an important factor driving water movement in the subsurface (Dingman, 2015; Moore et al., 1991) and is used commonly to describe probabilistic soil moisture distributions (Grayson et al., 1997; Western et al., 1999; Western et al., 2004). In addition, topographic gradients are known to affect atmospheric conditions and the development of microclimates (Chen et al., 1999; Frey et al., 2016; Hoylman et al., 2019; Young et al., 1997). To approximate the summed effect of topographically driven hydrological processes that can occur in complex terrain, we extracted digital elevation models derived from the Shuttle Radar Topography Mission dataset (Farr et al., 2007) at a 30-m resolution and conducted terrain analysis over the domain. We selected the topographic position index (TPI; Weiss, 2001) to describe the distribution of topographic positions across the western United States. The TPI was calculated at a 30-m grid resolution using a 1,000-m circular focal radius to preserve hillslope to watershed scale gradients in topographic relief (Figure 1c). The TPI is calculated as:

$$TPI = z_0 - \overline{z}$$
$$\overline{z} = \frac{1}{n_A} \sum_{i \in A} z_i$$

where  $z_0$  is the elevation of a pixel,  $\overline{z}$  is the focal mean elevation of pixels in an area (A) defined as the area within a 1,000-m radius of pixel  $z_0$ , and  $n_A$  is the number of pixels in A. The TPI is a topographic metric, which considers the relative difference in elevation of a pixel with respect to the focal mean of the surrounding pixels (topographic position). Progressively negative TPI values represent toe-slope and valley bottom topographic positions (i.e., convergent topographic positions), while progressively positive TPI values represent sideslope and ridgeline positions (i.e., divergent topographic positions). The TPI has been used to describe landforms across complex terrain (Weiss, 2001), spatial variability of soil N<sub>2</sub>O and CO<sub>2</sub> fluxes (Arias-Navarro et al., 2017), and patterns of vegetation productivity, biomass, and forest structure across watersheds (Hoylman et al., 2018; Swetnam et al., 2017).

10.1029/2019GL085546

#### 2.4. Plant Functional Type

To remove unwanted land cover types from the analysis and determine how plant functional types related to ecosystem sensitivity, we used the 2015 annual MCD12Q1.006 MODIS Land Cover Type 5, Plant Functional Type scheme (Friedl & Sulla-Menashe, 2015) resampled to a 30 m using nearest neighbor procedures in GEE. We removed all cereal and broadleaf croplands, urban and built-up lands, permanent snow and ice, nonvegetated lands, and water bodies from the analysis. Therefore, the ecosystems we characterized in this analysis were composed of evergreen needleleaf trees, evergreen broadleaf trees, deciduous needleleaf trees, deciduous broadleaf trees, grasses, and shrublands.

#### 2.5. Statistical Analysis:

#### 2.5.1. Pixel-Specific Linear Regression

For each 30-m pixel across the western United States we calculated linear regressions between the climatic water deficit and NPP to describe the local productivity-climate relationship (regression slope:  $_2$ ) using gridded time series from 1986 to 2015 (Figures 2a–2c). We used linear regressions due to computational limitations in response to the very large number of pixel-specific regressions considered (30-year annual time series of NPP and deficit at 2,870,751,368 locations). These simple regressions do not capture potential nonlinear behaviors between vegetation growth, mortality, and the climatic water deficit. Statistical models were computed using GEE (Gorelick et al., 2017). We did not remove regressions with nonsignificant slope coefficients (i.e., *p*-values > 0.05) in order to preserve pixels with slopes close to 0 (insensitive pixels; see Figure 2b). Instead, to assess the performance of each linear regression error (Figure S1 in the supporting information). We then normalized the linear regression slopes (Figure 2c) to account for different distributions of NPP and deficit at each pixel by calculating standardized regression coefficients (\*; Bring, 1994):

$$\beta^* = \beta_2 \times (\sigma_{\text{deficit}} / \sigma_{\text{NPP}})$$

where <sup>\*</sup> is the standardized regression slope (Figure 2d, considered ecosystem sensitivity), <sub>2</sub> is the original regression slope (Figure 2c),  $\sigma_{deficit}$  is the standard deviation of the pixel-specific deficit distribution, and  $\sigma_{NPP}$  is the standard deviation of the pixel-specific NPP distribution. This normalization was necessary to conduct cross-pixel comparisons of regression slopes (for example for bordering pixels with differing vegetation types). Vegetation type, density, and cover directly influence the NPP distribution of a pixel and thus, the raw regression slope. By accounting for spatial variation in NPP (and deficit) magnitude and variance we can compare ecosystem sensitivity across locations with differing climatic and physiological characteristics.

#### 2.5.2. Regional Analysis

Spatiotemporal gradients in the climatic water balance exhibit a strong control over global vegetation distributions (Stephenson, 1998; Woodward, 1987) and serve as a biologically meaningful representation of the climatic drivers of productivity (Vicente-Serrano et al., 2015). In order to quantify the effect of regional climate gradients on average ecosystem NPP and the sensitivity of ecosystem NPP to the annual water balance (ecosystem sensitivity; \*), we aggregated NPP and ecosystem sensitivity values for the western United States into 100 bins that were determined by the 30-year mean climatic water deficit. Each bin consisted of 1% of the domain area (100 deficit quantiles, ~25,836 km<sup>2</sup> in each bin) from which the median value, 5th and 95th quantiles of NPP, and ecosystem sensitivity (\*) were calculated. We compared these measures of response to the median deficit value for a deficit bin (Figures 1d and 3a). To assess how spatial patterns of dominant plant functional types varied within the regional continuum of ecosystem sensitivity, we calculated the climatic conditions which were associated with each plant functional type. To do so, we calculated the median (Figures 1d and 3a; black scatter point) and 5th and 95th deficit quantiles (Figures 1d and 3a; horizontal bars) experienced by each plant functional type across the western United States.

#### 2.5.3. Topographic Position Analysis

We evaluated how spatial patterns of NPP and ecosystem sensitivity (\*) were affected by hillslope scale gradients in topographic position by extracting NPP, ecosystem sensitivity, and climatic water deficit grids that were aligned with discrete classes of the TPI. Classes of TPI were determined using equal 20-m



**Figure 3.** (a) The relationship between the climatic water deficit (deficit = potential evapotranspiration – actual evapotranspiration) and ecosystem sensitivity ( $\beta^*$ ) for the western United States. We discretized the entire western United States into 100 quantiles of the deficit. For each deficit quantile the median (black line) and 5th and 95th quantiles (grey ribbon) of ecosystem sensitivities are shown. The effect of topography was quantified by repeating this analysis on discrete deficit regions defined by the TPI (lines color coded by red-blue color gradient). The lines transitioning from red to yellow to blue indicate a gradient from divergent to flat to convergent landscape positions. The 5th and 95th deficit quantiles associated with each plant functional type are represented by the horizontal bar, and the median value is represented by the scatter point on each horizontal bar. (b) The relative influence of topography on ecosystem sensitivities was calculated as the difference in ecosystem sensitivity ( $\beta^*$ ) between flat, highly convergent and highly divergent positions within a given deficit region.

breaks that ranged from TPI values of less than -190 m to greater than 190 m. Once we extracted grids associated with each TPI class, we replicated the regional analysis described above for each TPI class; we aggregated the data into 100 quantile bins based on the TPI-specific climatic water deficit distribution. Within each TPI-specific deficit bin, we calculated the median NPP and ecosystem sensitivity (\*) value (Figures 1d and 3a).

To quantify the relative effect of topography on ecosystem sensitivity across the western United States deficit gradient, we calculated the difference in ecosystem sensitivity ( $\Delta^*$ ; Figure 3b) between the flat TPI class (10 > TPI > -10; Figure 3a; yellow line) and the most convergent/divergent TPI class (Figures 3a and 3b; -190 > TPI, 190 < TPI; dark blue, dark red lines respectively). To do so, we interpolated these two lines into equal intervals of the deficit using a spline function and calculated the difference between the convergent/divergent and flat median sensitivity values across the deficit gradient.

#### 3. Results

The complexity of ecosystem climate sensitivity across the western United States was strongly influenced by the spatial alignment of the climatic water balance and local hillslope topography (Figures 3a and 3b). Across most regions of the west, we found a reduced sensitivity of NPP to annual climate fluctuations (ecosystem sensitivity) and greater rates of NPP across sites with convergent topography (Figures 1d and 3a and 3b). We also observed a reduction in ecosystem sensitivity along divergent hillslope positions, although rates of NPP in these locations were substantially lower than convergent sites (Figures 1d and 3a and 3b). Importantly, the effect of topography on ecosystem productivity was contingent on local aridity. Semiarid ecosystem productivity (absolute rate and sensitivity) was the most responsive to topographic effects (Figures 3a and 3b; 0.6 m < deficit < 1.0 m), and these climatic regions represent 43.8% of the western United States landscape. Here, vegetation in flat topographic positions was the most sensitive to annual fluctuations in the climatic water balance  $(10 > TPI > -10; 640,715 \text{ km}^2)$ , vegetation in divergent positions was less sensitive (TPI > 100; ~9,738 km<sup>2</sup>), and vegetation in convergent positions was the least sensitive (TPI < -100; ~8,126 km<sup>2</sup>; Figures 3a and 3b). The topographic effect was substantially lower in mesic/hygric (deficit < 0.6 m) or arid sites (1.0 m < deficit < 1.5 m). Finally, topographic position had little influence on ecosystem climate sensitivity in xeric locations with average deficits >1.5m, although divergent positions tended to be slightly less sensitive than the regional average.

The majority of the western United States exhibited negative ecosystem climate sensitivities, representing reduced NPP during dry years and water-limited ecosystem response. Water-limited conditions tended to occur where the annual climatic water deficit was greater than 0.3 m (Figures 1b, 2d, and 3a). Hygric areas with deficits less than 0.3 m were aligned with positive ecosystem sensitivities (Figure 3a), representing increasing NPP with increasing annual deficit. These areas were generally located in mountainous terrain (e.g., Sierra Nevada Range and Rocky Mountain Range) and in coastal climatic regions (e.g., Washington/Oregon coast and the Olympic Peninsula) characterized by energy-limited ecosystems. Ecosystems were the most sensitive to annual variations in the climatic water balance in regions with deficits of ~1.5 m; however, in xeric climates (deficit > 1.5 m), ecosystem-climate sensitivity decreased (\* trended toward 0).

Vegetation land cover type was related to the 30-year mean deficit of a pixel (Figure 3a). Generally forest ecosystems were present in regions with less than ~1-m deficit, while grasslands were present from ~0.5- to 1.3-m deficit and shrublands were present from ~1.1- to 1.8-m deficit (Figure 3a). Evergreen needleleaf and evergreen broadleaf land cover types dominated ecosystems in the most hygric portions of the western United States and exhibited the most positive ecosystem sensitivity to the deficit. The most negative ecosystem sensitivity occurred at the transition from grass to shrub dominated landscapes where the deficit was 1.3-1.5m.

#### 4. Discussion

There has been a need to characterize "where and when, across the diverse and dynamic environments of the globe" we can expect hydrological processes associated with topography (e.g., groundwater, shallow soil moisture, and microclimates) to be relevant for predictions of the carbon balance (Fan et al., 2019; McLaughlin et al., 2017). We found that vegetation in convergent zones were generally the least sensitive to climate (in water-limited settings) or had the greatest positive response to dryness (in energy-limited settings) across the majority of the western United States (Figure 3a). However, the relative importance of topography for ecosystem sensitivity varied across gradients in the climatic water balance. Our analysis provides strong evidence that topography has the greatest effect on ecosystem sensitivity in semiarid climates but little effect in especially wet or dry regions (Figure 3b). To our knowledge, this finding has never been quantified and reported at this scale or level of detail. Our results also highlight that vegetation in divergent topographic positions are less sensitive to climatic variability, albeit to a lesser degree than vegetation in convergent positions (Figures 3a and 3b). Vegetation in flat positions was the most sensitive to climate, except in energy-limited regions where the climatic water deficit was low (deficit < 0.3).

Our results reflect the strong gradients in moisture associated with topographic positions that buffer ecosystems from regional climate dynamics (Hoylman et al., 2018; Hoylman et al., 2019; Ivanov et al., 2008;

Swetnam et al., 2017). We posit that semiarid regions experience transient to prolonged shallow subsurface flow (e.g., Jencso et al., 2009) in convergent locations, thereby enhancing soil moisture and reducing ecosystem water limitations along downslope positions (Hawthorne & Miniat, 2018; Hwang et al., 2012; Maxwell & Condon, 2016). Atmospheric microclimates also occur in convergent landscape positions and contribute to feedbacks that reduce evaporative fluxes from the soil (Entekhabi et al., 1996; Hoylman et al., 2019) and prolong the buffering effect produced by soil moisture and groundwater movement in the near surface rooting zone. In these locations, enhanced soil moisture buffers ecosystem water use (i.e., transpiration) from climate (Hawthorne & Miniat, 2018) and prolongs the growing season in more convergent hillslope positions (Hoylman et al., 2018; Hoylman et al., 2019; Martin et al., 2017). Weathered bedrock has also been shown to be a significant moisture storage reservoir, especially along upslope divergent positions (Rempe & Dietrich, 2018). Morphological adaptation of vegetation along divergent positions may exploit this moisture by extending roots deep into the weathered bedrock (Fan et al., 2017). Thus, "rock moisture" may sustain plant water demand during drought and contribute to reduced ecosystem sensitivity on divergent hillslope positions (Figures 3a and 3b).

The relatively small percentage of the western United States ecosystem with positive slopes (12.4%; Figures 2d, 3a, and S2a) are indicative of locations where energy availability (radiant energy, temperature, growing season length, and thus annual PET) as opposed to water availability was a dominant climatic driver of ecosystem productivity (Schuur, 2003; Churkina et al., 2005; Seddon et al., 2016; Figure 3a). In these climatic regions, flatter areas had higher rates of productivity compared to divergent or convergent hillslope positions (Figure 1d), likely due to greater exposure and therefore daily photosynthetically active radiation. The topographic effect on ecosystem sensitivity was reduced in this energy-limited environment due to more homogeneous moisture availability across all landscape positions (Figures 3a and 3b). Positive relationships between NPP and the deficit (i.e., ecosystem sensitivity values greater than 0 across the deficit gradient; dashed line in Figure 3a) may also reflect ecosystem water usage from near-surface reservoirs such as persistent regional groundwater tables (Fan et al., 2013; Fan et al., 2017). Biogeochemical and biological factors (such as nutrients, biological competition, and species richness; Mittelbach et al., 2001; LeBauer & Treseder, 2008) also contribute to spatial patterns of NPP and the associated sensitivity to climate, especially where water limitations do not dominate. These energy-limited ecosystems may become a greater sink for atmospheric carbon as climate increases annual PET across the west (Fu & Feng, 2014).

The dominance of negative ecosystem sensitivities across the majority of the western United States (i.e., deficit > 0.3 m; Figure 3a; 87.6% of the land area) agrees with previous work and provides further evidence that water is a significant limitation for ecosystem productivity across the region (Littell et al., 2008; Seddon et al., 2016; Vicente-Serrano et al., 2013). In general, ecosystems became more sensitive to annual climatic fluctuations (\* decreased) as the climatic water deficit became more arid and the deficit increased (Figure 3a). The increasing sensitivity of NPP to the climatic water balance indicates that these ecosystems become increasingly opportunistic[vulnerable] to moisture availability[limitation] as they are located in progressively drier climates. Importantly, in moderately water-limited regions (semiarid), vegetation in zones of topographic convergence had the highest rates of NPP (Figure 1d) and were the least sensitive to variability in the climatic water balance (Figure 3a).

Ecosystem-climate sensitivity decreased (\* trended toward 0) where shrubs became the dominant plant functional type over grasses (Figure 3a) in xeric climates (deficit >1.5 m). This inflection may reflect species traits more adapted to arid environmental conditions and rapidly changing water availability (Chaves et al., 2003), although NPP in these regions remained very low (Figure 1d). Drought tolerance has been observed in shrubs and can be attributed to seasonal modification to leaf water relations and gas exchange capacity that result in positive leaf turgor in times of extreme soil water deficits (Reynolds et al., 1999). Shrub drought avoidance has also been observed in species with deep rooting depths, preventing prolonged stomatal closure during drought (Hinckley et al., 1983).

Analysis of ecosystem sensitivity to climate can identify those ecosystems vulnerable to water stress, providing a context for identifying areas more likely to experience drought-induced declines in productivity and species shifts. As regional climatic water deficits become progressively more arid throughout the century (Barnett et al., 2005; Ficklin & Novick, 2017; Fu & Feng, 2014; IPCC, 2013; Leung et al., 2004) we expect the climate sensitivity of western United States ecosystems to increase, resulting in greater vulnerability to drought induced stress and decreased NPP (e.g., shifting right along Figures 1d and 3a). Exceptions to this generality may arise in hygric or xeric regions where significant moisture surpluses (hygric) and physiological adaptations (xeric) will likely slow ecosystem responses to a changing water balance.

#### 5. Limitations and Implications

Our results show a clear effect of local landscape topography and the importance of its distribution within a given region for quantifying ecosystem sensitivity to drought (Figures 3a and 3b). We did not explicitly characterize the active hydrological processes that occurred across varying topographic settings of the western United States climate gradient. Instead, we inferred key processes from the wealth of experimental and theoretical hydrology studies that have documented the importance of landscape topography for water redistribution and storage in the subsurface and near-surface atmosphere (Beven & Kirkby, 1979; Hoylman et al., 2019; Jencso et al., 2009; Rasmussen et al., 2015; Troch et al., 2003; Western et al., 2004), contributing to the persistence of plant available water in convergent settings (Fan et al., 2017; Hawthorne & Miniat, 2018; Maxwell & Condon, 2016; Moeslund et al., 2013; Tai et al., 2017) and in some cases, divergent settings (Fan et al., 2017; Rempe & Dietrich, 2018).

Hydrological processes and their persistence are highly dependent upon the topography, soil, geology, vegetation, anthropogenic modification, and the amount of water and energy available in a given landscape (Wagener et al., 2007). In the future, it will be important for ecological studies to explicitly identify and incorporate the spatial distribution of active hillslope hydrology processes, and their potential to change with climate shifts, if we are to fully predict ecosystem trajectories in water-limited landscapes (Fan et al., 2019; Simeone et al., 2018; Tai et al., 2017). One potential way forward is to perform site-specific analysis across the ecosystem sensitivity gradient (Figures 2d and 3a) using field-based hydrology approaches (e.g., Martin et al., 2017; Hawthorne & Miniat, 2018; Hoylman et al., 2019) or distributed ecohydrology modeling frameworks that couple the water and carbon cycles (e.g., Maneta & Silverman, 2013; Simeone et al., 2018; Tague & Band, 2004)

We present a simple and mappable empirical method to identify ecosystems that are buffered from annual climate fluctuations (e.g., Figure S2a), an important consideration when describing local ecosystem-climate relationships and for identifying conservation and management priorities across scales (Morelli et al., 2016). It is well known that factors such as climate and climate change, combined with topography, soil types, nutrient availability, and disturbance regimes, are central drivers of the state and potential trajectory of ecosystems (Box, 1996; Delcourt et al., 1982; McCarty, 2001). Ultimately, plant species must "migrate" in response to climate change, as they can only survive, compete, and reproduce within the range of climate and resource availability to which they are physiologically adapted. By combining gridded climatic water balance data with measures of biological response (i.e., ecosystem NPP) we can assess the potential for plant communities to persist across differing landscape positions into the future. While the zones of resistant vegetation identified here represent a small fraction of the landscape (Figure S2a), their persistence through future droughts effects the greater ecosystem species assemblages outside of their regional distribution (Dobrowski, 2011; McLaughlin et al., 2017; Rull, 2009) and influence the migration corridors of ecotones as PET and AET partitioning changes into the future (Allen & Breshears, 1998).

Decision-makers often have to prioritize options for action on the ground. Site-specific analyses that map vegetation sensitivity (Figures 2d and S2a) allow for better prioritization, given limited resources. Our approach takes a first step toward classification of sensitive ecosystems across the entirety of the western United States using simple linear models. The root-mean-square error of the linear regressions tended to be larger in areas with high rates of NPP, which generally align with areas with positive slopes (Figures 2d and S1) and energy limitations. For site-specific analysis it would be possible to incorporate more complex statistical models that would better characterize potential nonlinear behaviors between productivity and water availability due to physiological differences across plant types, where the climatic water balance is not a significant constraint on NPP, or where mortality events occur in response to extreme events. Importantly, it is possible to overlay maps of ecosystem sensitivity with maps of other features, vulnerabilities, management potential, historical legacies, and land ownership (Frazier et al., 2014; Morelli et al., 2016). Indices of ecosystem sensitivity/vulnerability should not be dependent upon generalized characteristics of the ecosystem, but stressor-specific environmental indicators that include information on local

exposure pathways, ecosystem sensitivity, and the potential adaptive capacity (Beroya-Eitner, 2016; De Lange et al., 2010). Following identification of conservation or management targets, improved management options may be identified.

#### 6. Conclusion

Our results show that hillslope topography exerts a strong control over spatial patterns of primary productivity and ecosystem sensitivity to climate. The relative effect of topographic convergence and divergence on ecosystem productivity varied across the western United States climate gradient. Under water-limited conditions, vegetation in convergent terrain was generally less sensitive to climate variability and more productive than vegetation in flat and divergent landscape positions, especially in semiarid environments. Vegetation in divergent terrain was also less sensitive to climate than vegetation in flat positions, but comparatively less productive and more sensitive than vegetation in convergent terrain. Conversely, topography had little effect on ecosystem sensitivity to climate in hygric and xeric environments where climatic drivers were extreme (e.g., larger moisture surplus or deficit respectively). This work provides a mappable approach for regional planning to maximize conservation in landscapes more resistant to perturbations.

#### References

- Abatzoglou, J. T. (2013). Development of gridded surface meteorological data for ecological applications and modelling. *International Journal of Climatology*, 33, 121–131.
- Allen, C. D., & Breshears, D. D. (1998). Drought-induced shift of a forest-woodland ecotone: rapid landscape response to climate variation. *Proceedings of the National Academy of Sciences*, 95(25), 14,839–14,842.
- Arias-Navarro, C., Díaz-Pinés, E., Klatt, S., Brandt, P., Rufino, M. C., Butterbach-Bahl, K., & Verchot, L. V. (2017). Spatial variability of soil N<sub>2</sub>O and CO<sub>2</sub> fluxes in different topographic positions in a tropical montane forest in Kenya. *Journal of Geophysical Research: Biogeosciences*, 122, 514–527. https://doi.org/10.1002/2016JG003667
- Ashcroft, M. B. (2010). Identifying refugia from climate change. Journal of Biogeography, 37(8), 1407-1413.
- Barnett, T. P., Adam, J. C., & Lettenmaier, D. P. (2005). Potential impacts of a warming climate on water availability in snow-dominated regions. *Nature*, 438(7066), 303.
- Beroya-Eitner, M. A. (2016). Ecological vulnerability indicators. Ecological indicators, 60, 329-334.
- Beven, K. J., & Kirkby, M. J. (1979). A physically based, variable contributing area model of basin hydrology/Un modèle à base physique de zone d'appel variable de l'hydrologie du bassin versant. *Hydrological Sciences Journal*, 24(1), 43–69.
- Box, E. O. (1996). Plant functional types and climate at the global scale. *Journal of Vegetation Science*, 7(3), 309–320.
- Brubaker, S. C., Jones, A. J., Lewis, D. T., & Frank, K. (1993). Soil properties associated with landscape position. Soil Science Society of America Journal, 57(1), 235–239.
- Carroll, C., Parks, S. A., Dobrowski, S. Z., & Roberts, D. R. (2018). Climatic, topographic, and anthropogenic factors determine connectivity between current and future climate analogs in North America. *Global Change Biology*, 24(11), 5318–5331. https://doi.org/10.1111/gcb.14373
- Chaves, M. M., Maroco, J. P., & Pereira, J. S. (2003). Understanding plant responses to drought—from genes to the whole plant. Functional plant biology, 30(3), 239–264.
- Chen, J., Saunders, S. C., Crow, T. R., Naiman, R. J., Brosofske, K. D., Mroz, G. D., et al. (1999). Microclimate in forest ecosystem and landscape ecology: variations in local climate can be used to monitor and compare the effects of different management regimes. *BioScience*, 49(4), 288–297.
- Churkina, G., & Running, S. W. (1998). Contrasting climatic controls on the estimated productivity of global terrestrial biomes. *Ecosystems*, *1*(2), 206–215.
- Churkina, G., Schimel, D., Braswell, B. H., & Xiao, X. (2005). Spatial analysis of growing season length control over net ecosystem exchange. *Global Change Biology*, *11*(10), 1777–1787.
- Clair, J. S., Moon, S., Holbrook, W. S., Perron, J. T., Riebe, C. S., Martel, S. J., et al. (2015). Geophysical imaging reveals topographic stress control of bedrock weathering. *Science*, 350(6260), 534–538. https://doi.org/10.1126/science.aab2210
- De Lange, H. J., Sala, S., Vighi, M., & Faber, J. H. (2010). Ecological vulnerability in risk assessment—A review and perspectives. Science of the Total Environment, 408(18), 3871–3879.
- Delcourt, H. R., Delcourt, P. A., & Webb, T. III (1982). Dynamic plant ecology: The spectrum of vegetational change in space and time. Quaternary Science Reviews, 1(3), 153–175.
- Dietrich, W. E., & Perron, J. T. (2006). The search for a topographic signature of life. Nature, 439(7075), 411.
- Dingman, S. L. (2015). Physical hydrology. Long Grove, IL: Waveland press.
- Dobrowski, S. Z. (2011). A climatic basis for microrefugia: The influence of terrain on climate. *Global Change Biology*, 17(2), 1022–1035.
  Dobrowski, S. Z., Abatzoglou, J., Swanson, A. K., Greenberg, J. A., Mynsberge, A. R., Holden, Z. A., & Schwartz, M. K. (2013). The climate velocity of the contiguous United States during the 20th century. *Global Change Biology*, 19(1), 241–251.
- Dobrowski, S. Z., & Parks, S. A. (2016). Climate change velocity underestimates climate change exposure in mountainous regions. Nature Communications, 7, 12349.
- Eagleson, P. S. (1978). Climate, soil, and vegetation: 6. Dynamics of the annual water balance. Water Resources Research, 14(5), 749–764.
  Entekhabi, D., Rodriguez-Iturbe, I., & Castelli, F. (1996). Mutual interaction of soil moisture state and atmospheric processes. Journal of Hydrology, 184(1-2), 3–17.
- Fan, Y., Clark, M., Lawrence, D. M., Swenson, S., Band, L. E., Brantley, S. L., et al. (2019). Hillslope hydrology in global change research and Earth system modeling. *Water Resources Research*, 55, 1737–1772. https://doi.org/10.1029/2018WR023903
- Fan, Y., Li, H., & Miguez-Macho, G. (2013). Global patterns of groundwater table depth. Science, 339(6122), 940–943. https://doi.org/ 10.1126/science.1229881

#### Acknowledgments

This work was made possible by a USDA NIFA McIntire Stennis award 233327 to Jencso, NSF grants DEB-1457749 and DEB-1457720 to Jencso and Hu, and a NASA applied science program Wildland Fire award (agreement NNH11ZDA001N-FIRES) to Holden. The authors also thank NSF EPSCoR Track-1 EPS-1101342 (INSTEP 3) for support. Data for this analysis are available at https://doi.org/10.5281/ zenodo.3460768.

- Fan, Y., Miguez-Macho, G., Jobbágy, E. G., Jackson, R. B., & Otero-Casal, C. (2017). Hydrologic regulation of plant rooting depth. Proceedings of the National Academy of Sciences, 114(40), 10,572–10,577.
- Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., et al. (2007). The shuttle radar topography mission. Reviews of Geophysics, 45, RG2004. https://doi.org/10.1029/2005RG000183

Feng, M., Huang, C., Channan, S., Vermote, E. F., Masek, J. G., & Townshend, J. R. (2012). Quality assessment of Landsat surface reflectance products using MODIS data. *Computers & Geosciences*, 38(1), 9–22.

- Ficklin, D. L., & Novick, K. A. (2017). Historic and projected changes in vapor pressure deficit suggest a continental-scale drying of the United States atmosphere. *Journal of Geophysical Research: Atmospheres*, 122, 2061–2079. https://doi.org/10.1002/ 2016JD025855
- Frazier, T. G., Thompson, C. M., & Dezzani, R. J. (2014). A framework for the development of the SERV model: A spatially explicit resilience-vulnerability model. Applied Geography, 51, 158–172.
- Frey, S. J., Hadley, A. S., Johnson, S. L., Schulze, M., Jones, J. A., & Betts, M. G. (2016). Spatial models reveal the microclimatic buffering capacity of old-growth forests. *Science Advances*, 2(4), e1501392.
- Friedl, M., Sulla-Menashe, D. (2015). MCD12Q1 MODIS/Terra+Aqua Land Cover Type Yearly L3 Global 500m SIN Grid V006 [Data set]. NASA EOSDIS Land Processes DAAC. doi: https://doi.org/10.5067/MODIS/MCD12Q1.006
- Fry, J., Xian, G. Z., Jin, S., Dewitz, J., Homer, C. G., Yang, L., et al. (2011). Completion of the 2006 National Land Cover Database for the conterminous United States. *Photogrammetric Engineering and Remote Sensing*, 77(9), 858–864.

Fu, Q., & Feng, S. (2014). Responses of terrestrial aridity to global warming. Journal of Geophysical Research: Atmospheres, 119, 7863–7875. https://doi.org/10.1002/2014JD021608

Gabet, E. J., Edelman, R., & Langner, H. (2006). Hydrological controls on chemical weathering rates at the soil-bedrock interface. *Geology*, 34(12), 1065–1068.

Gorelick, N., Hancher, M., Dixon, M., Ilyushchenko, S., Thau, D., & Moore, R. (2017). Google Earth Engine: Planetary-scale geospatial analysis for everyone. *Remote Sensing of Environment*, 202, 18–27. https://doi.org/10.1016/j.rse.2017.06.031

- Grayson, R. B., Western, A. W., Chiew, F. H., & Blöschl, G. (1997). Preferred states in spatial soil moisture patterns: Local and nonlocal controls. Water Resources Research, 33(12), 2897–2908.
- Hawthorne, S., & Miniat, C. F. (2018). Topography may mitigate drought effects on vegetation along a hillslope gradient. *Ecohydrology*, *11*(1), e1825.
- Hinckley, T. M., Duhme, F., Hinckley, A. R., & Richter, H. (1983). Drought relations of shrub species: Assessment of the mechanisms of drought resistance. *Oecologia*, 59(2-3), 344–350.

Holden, Z. A., Jolly, W. M., Swanson, A., Warren, D. A., Jencso, K., Maneta, M., et al. (2019). TOPOFIRE: A topographically resolved wildfire danger and drought monitoring system for the conterminous United States. *Bulletin of the American Meteorological Society*, 100(9), 1607–1613. https://doi.org/10.1175/BAMS-D-18-0178.1

Homer, C., Dewitz, J., Fry, J., Coan, M., Hossain, N., Larson, C., et al. (2007). Completion of the 2001 National Land Cover Database for the conterminous United States. *Photogrammetric engineering and remote sensing*, 73(4), 337.

- Homer, C., Dewitz, J., Yang, L., Jin, S., Danielson, P., Xian, G., et al. (2015). Completion of the 2011 National Land Cover Database for the conterminous United States—Representing a decade of land cover change information. *Photogrammetric Engineering & Remote Sensing*, 81(5), 345–354.
- Hoylman, Z. H., Jencso, K. G., Hu, J., Holden, Z. A., Martin, J. T., & Gardner, W. P. (2019). The climatic water balance and topography control spatial patterns of atmospheric demand, soil moisture, and shallow subsurface flow. *Water Resources Research*, 55, 2370–2389. https://doi.org/10.1029/2018WR023302
- Hoylman, Z. H., Jencso, K. G., Hu, J., Martin, J. T., Holden, Z. A., Seielstad, C. A., & Rowell, E. M. (2018). Hillslope topography mediates spatial patterns of ecosystem sensitivity to climate. *Journal of Geophysical Research: Biogeosciences*, 123, 353–371. https://doi.org/ 10.1002/2017JG004108
- Hwang, T., Band, L. E., Vose, J. M., & Tague, C. (2012). Ecosystem processes at the watershed scale: Hydrologic vegetation gradient as an indicator for lateral hydrologic connectivity of headwater catchments. *Water Resources Research*, 48, W06514. https://doi.org/10.1029/ 2011WR011301
- IPCC (2013). Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 1535 pp. https://doi.org/10.1017/CB09781107415324
- Istanbulluoglu, E., Yetemen, O., Vivoni, E. R., Gutiérrez-Jurado, H. A., & Bras, R. L. (2008). Eco-geomorphic implications of hillslope aspect: Inferences from analysis of landscape morphology in central New Mexico. *Geophysical Research Letters*, 35, L14403. https://doi. org/10.1029/2008GL034477
- Ivanov, V. Y., Bras, R. L., & Vivoni, E. R. (2008). Vegetation-hydrology dynamics in complex terrain of semiarid areas: 2. Energy-water controls of vegetation spatiotemporal dynamics and topographic niches of favorability. Water Resources Research, 44, W03430. https://doi.org/10.1029/2006WR005595
- Jencso, K. G., McGlynn, B. L., Gooseff, M. N., Wondzell, S. M., Bencala, K. E., & Marshall, L. A. (2009). Hydrologic connectivity between landscapes and streams: Transferring reach-and plot-scale understanding to the catchment scale. Water Resources Research, 45, W04428. https://doi.org/10.1029/2008WR007225
- Keppel, G., Mokany, K., Wardell-Johnson, G. W., Phillips, B. L., Welbergen, J. A., & Reside, A. E. (2015). The capacity of refugia for conservation planning under climate change. Frontiers in Ecology and the Environment, 13(2), 106–112.
- LeBauer, D. S., & Treseder, K. K. (2008). Nitrogen limitation of net primary productivity in terrestrial ecosystems is globally distributed. *Ecology*, 89(2), 371–379.

Leung, L. R., Qian, Y., Bian, X., Washington, W. M., Han, J., & Roads, J. O. (2004). Mid-century ensemble regional climate change scenarios for the western United States. *Climatic Change*, 62(1-3), 75–113.

Littell, J. S., Peterson, D. L., & Tjoelker, M. (2008). Douglas-fir growth in mountain ecosystems: water limits tree growth from stand to region. *Ecological Monographs*, 78(3), 349–368.

- Maneta, M. P., & Silverman, N. L. (2013). A spatially distributed model to simulate water, energy, and vegetation dynamics using information from regional climate models. *Earth Interactions*, 17(11), 1–44.
- Martin, J., Looker, N., Hoylman, Z., Jencso, K., & Hu, J. (2017). Hydrometeorology organizes intra-annual patterns of tree growth across time, space and species in a montane watershed. New Phytologist, 215(4), 1387–1398.

Masek, J. G., Vermote, E. F., Saleous, N. E., Wolfe, R., Hall, F. G., Huemmrich, K. F., et al. (2006). A Landsat surface reflectance dataset for North America, 1990-2000. IEEE Geoscience and Remote Sensing Letters, 3(1), 68–72.

Maxwell, R. M., & Condon, L. E. (2016). Connections between groundwater flow and transpiration partitioning. *Science*, 353(6297), 377–380.

McCarty, J. P. (2001). Ecological consequences of recent climate change. Conservation biology, 15(2), 320-331.

McLaughlin, B. C., Ackerly, D. D., Klos, P. Z., Natali, J., Dawson, T. E., & Thompson, S. E. (2017). Hydrologic refugia, plants, and climate change. Global Change Biology, 23(8), 2941–2961.

Mittelbach, G. G., Steiner, C. F., Scheiner, S. M., Gross, K. L., Reynolds, H. L., Waide, R. B., et al. (2001). What is the observed relationship between species richness and productivity? *Ecology*, 82(9), 2381–2396.

- Moeslund, J. E., Arge, L., Bøcher, P. K., Dalgaard, T., Ejrnæs, R., Odgaard, M. V., & Svenning, J. C. (2013). Topographically controlled soil moisture drives plant diversity patterns within grasslands. *Biodiversity and conservation*, 22(10), 2151–2166.
- Moore, I. D., Grayson, R. B., & Ladson, A. R. (1991). Digital terrain modelling: A review of hydrological, geomorphological, and biological applications. *Hydrological processes*, 5(1), 3–30.
- Morelli, T. L., Daly, C., Dobrowski, S. Z., Dulen, D. M., Ebersole, J. L., Jackson, S. T., et al. (2016). Managing climate change refugia for climate adaptation. *PLoS One*, 11(8), e0159909.

Rasmussen, C., Pelletier, J. D., Troch, P. A., Swetnam, T. L., & Chorover, J. (2015). Quantifying topographic and vegetation effects on the transfer of energy and mass to the critical zone. *Vadose Zone Journal*, 14(11).

Rempe, D. M., & Dietrich, W. E. (2018). Direct observations of rock moisture, a hidden component of the hydrologic cycle. Proceedings of the National Academy of Sciences, 115(11), 2664–2669.

Reynolds, J. F., Virginia, R. A., Kemp, P. R., De Soyza, A. G., & Tremmel, D. C. (1999). Impact of drought on desert shrubs: Effects of seasonality and degree of resource island development. *Ecological Monographs*, 69(1), 69–106.

Robinson, N. P., Allred, B. W., Smith, W. K., Jones, M. O., Moreno, A., Erickson, T. A., et al. (2018). Terrestrial primary production for the conterminous United States derived from Landsat 30 m and MODIS 250 m. *Remote Sensing in Ecology and Conservation*, 4(3), 264–280. https://doi.org/10.1002/rse2.74

Rodriguez-Iturbe, I. (2000). Ecohydrology: A hydrologic perspective of climate-soil-vegetation dynamies. *Water Resources Research*, 36(1), 3–9.

- Rull, V. (2009). Microrefugia. Journal of Biogeography, 36(3), 481-484.
- Running, S. W., Nemani, R. R., Heinsch, F. A., Zhao, M., Reeves, M., & Hashimoto, H. (2004). A continuous satellite-derived measure of global terrestrial primary production. AIBS Bulletin, 54(6), 547–560.

Schuur, E. A. (2003). Productivity and global climate revisited: The sensitivity of tropical forest growth to precipitation. *Ecology*, 84(5), 1165–1170.

Seddon, A. W., Macias-Fauria, M., Long, P. R., Benz, D., & Willis, K. J. (2016). Sensitivity of global terrestrial ecosystems to climate variability. *Nature*, 531(7593), 229.

Seibert, J., Stendahl, J., & Sørensen, R. (2007). Topographical influences on soil properties in boreal forests. *Geoderma*, 141(1-2), 139–148.

- Simeone, C., Maneta, M. P., Holden, Z. A., Sapes, G., Sala, A., & Dobrowski, S. Z. (2018). Coupled ecohydrology and plant hydraulics modeling predicts ponderosa pine seedling mortality and lower treeline in the US Northern Rocky Mountains. *New Phytologist*, *10.*
- Sims, D. A., Rahman, A. F., Cordova, V. D., El-Masri, B. Z., Baldocchi, D. D., Bolstad, P. V., et al. (2008). A new model of gross primary productivity for North American ecosystems based solely on the enhanced vegetation index and land surface temperature from MODIS. *Remote Sensing of Environment*, 112(4), 1633–1646.
- Smith, W. K., Reed, S. C., Cleveland, C. C., Ballantyne, A. P., Anderegg, W. R., Wieder, W. R., et al. (2016). Large divergence of satellite and Earth system model estimates of global terrestrial CO 2 fertilization. *Nature Climate Change*, 6(3), 306.
- Stephenson, N. (1998). Actual evapotranspiration and deficit: Biologically meaningful correlates of vegetation distribution across spatial scales. *Journal of Biogeography*, 25(5), 855–870.
- Swetnam, T. L., Brooks, P. D., Barnard, H. R., Harpold, A. A., & Gallo, E. L. (2017). Topographically driven differences in energy and water constrain climatic control on forest carbon sequestration. *Ecosphere*, 8(4).

Tague, C. L., & Band, L. E. (2004). RHESSys: Regional Hydro-Ecologic Simulation System—An object-oriented approach to spatially distributed modeling of carbon, water, and nutrient cycling. *Earth interactions*, 8(19), 1–42.

Tai, X., Mackay, D. S., Anderegg, W. R., Sperry, J. S., & Brooks, P. D. (2017). Plant hydraulics improves and topography mediates prediction of aspen mortality in southwestern USA. New Phytologist, 213(1), 113–127.

Troch, P. A., Paniconi, C., & Emiel van Loon, A. E. (2003). Hillslope-storage Boussinesq model for subsurface flow and variable source areas along complex hillslopes: 1. Formulation and characteristic response. Water Resources Research, 39(11), 1316. https://doi.org/10.1029/ 2002WR001728

Vermote, E., Justice, C., Claverie, M., & Franch, B. (2016). Preliminary analysis of the performance of the Landsat 8/OLI land surface reflectance product. *Remote Sensing of Environment*, 185, 46–56.

Vicente-Serrano, S. M., Camarero, J. J., Zabalza, J., Sangüesa-Barreda, G., López-Moreno, J. I., & Tague, C. L. (2015). Evapotranspiration deficit controls net primary production and growth of silver fir: Implications for Circum-Mediterranean forests under forecasted warmer and drier conditions. Agricultural and forest meteorology, 206, 45–54.

Vicente-Serrano, S. M., Gouveia, C., Camarero, J. J., Beguería, S., Trigo, R., López-Moreno, J. I., et al. (2013). Response of vegetation to drought time-scales across global land biomes. Proceedings of the National Academy of Sciences, 110(1), 52–57.

Vicente-Serrano, S. M., Miralles, D. G., Domínguez-Castro, F., Azorin-Molina, C., El Kenawy, A., McVicar, T. R., et al. (2018). Global assessment of the Standardized Evapotranspiration Deficit Index (SEDI) for drought analysis and monitoring. *Journal of Climate*, 31(14), 5371–5393. https://doi.org/10.1175/JCLI-D-17-0775.1

Wagener, T., Sivapalan, M., Troch, P., & Woods, R. (2007). Catchment classification and hydrologic similarity. Geography Compass, 1(4), 901–931.

Weiss, A. (2001). Topographic position and landforms analysis. In Poster presentation, ESRI user conference, San Diego, CA (Vol. 200).Western, A. W., Grayson, R. B., Blöschl, G., Willgoose, G. R., & McMahon, T. A. (1999). Observed spatial organization of soil moisture and its relation to terrain indices. *Water Resources Research*, 35(3), 797–810.

Western, A. W., Zhou, S. L., Grayson, R. B., McMahon, T. A., Blöschl, G., & Wilson, D. J. (2004). Spatial correlation of soil moisture in small catchments and its relationship to dominant spatial hydrological processes. *Journal of Hydrology*, 286(1-4), 113–134. Woodward, F. I. (1987). Climate and plant distribution. New York, NY: Cambridge University Press.

Yetemen, O., Istanbulluoglu, E., & Vivoni, E. R. (2010). The implications of geology, soils, and vegetation on landscape morphology:

Inferences from semi-arid basins with complex vegetation patterns in Central New Mexico, USA. *Geomorphology*, *116*(3-4), 246–263. Young, K. L., Woo, M. K., & Edlund, S. A. (1997). Influence of local topography, soils, and vegetation on microclimate and hydrology at a high Arctic site, Ellesmere Island, Canada. *Arctic and Alpine Research*, *29*(3), 270–284.